Author's response to reviews

Title: Effect of a Herbal-Leucine mix on the IL-1beta-induced cartilage degradation and inflammatory gene expression in human chondrocytes

Authors:

Nahid Akhtar (nxa155@case.edu)
Mark JS Miller (markjsmo3@gmail.com)
Tariq M Haqqi (thaqqi@metrohealth.org)

Version: 5 Date: 18 August 2011

Author's response to reviews: see over
Please find the revised research article entitled “Effect of a Herbal-Leucine mix on the IL-1β-induced cartilage degradation and inflammatory gene expression in human chondrocytes” (MS: 9816785705512274) for favor of publication in BMC Complementary and Alternative Medicine. All the formatting changes have been made in the manuscript as suggested. The point-by-point description of changes made is as follows

**Major revisions**

Affiliations: Please write affiliation 2 in full.

**Authors’ response:** Please note that the second author (Mark JS Miller) is retired and has no institutional affiliation. The details of street, city, postal code, state and country has been provided in the manuscript.

**Minor revisions**

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

**Authors’ response:** Manuscript has been checked for the typo-graphical errors, final correction and revision.

Please feel free to contact me if you have any concern.

Sincerely yours,

Tariq M. Haqqi, Ph. D

Corresponding author

Department of Medicine/Rheumatology,
MetroHealth Medical Centre/Case Western Reserve University,
2500 Metro Health Drive, Cleveland, Ohio 44109.
Tel.: +1 216-778-5154; Fax: +1 216-778-8376. Email: thaqqi@metrohealth.org